Author's response to reviews

Title: A cluster randomised pragmatic trial applying Self-determination theory to type 2 diabetes care in general practice

Authors:

Lise Juul (lise.juul@alm.au.dk)
Helle T. Maindal (htm@alm.au.dk)
Vibeke Zoffmann (vzof@steno.dk)
Morten Frydenberg (morten@biostat.au.dk)
Anneli Sandbaek (annelli.sandbaek@alm.au.dk)

Version: 3 Date: 27 October 2011

Author's response to reviews:

Dear Tim Sands

Thank you for your response.

Description on the ethical approval of the trial has now been added to the manuscript:
"A cluster randomised pragmatic trial applying Self-determination theory to type 2 diabetes care in general practice".

Best regards,

Lise Juul